Author's response to reviews

Title: Contribution of main causes of death to social inequalities in mortality in the whole population of Scania, Sweden

Authors:

Maria Rosvall (Maria.rosvall@med.lu.se)
Basile Chaix (Basile.chaix@med.lu.se)
John Lynch (John.lynch@mcgill.ca)
Martin Lindstrom (Martin.lindstrom@med.lu.se)
Juan Merlo (Juan.merlo@med.lu.se)

Version: 6 Date: 20 March 2006

Author's response to reviews:

Title & Abstract - We have checked so these are identical in the manuscript and in the submission system.

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system. This has been made.

Section headings - We have removed all unnecessary capitals from the manuscript.

Website addresses - The website addresses are now cited in the reference list and removed from the manuscript text.

Highlighting/Tracking - Has been removed.

Figures - The figures are cropped as closely as possible